

Received 23 July 2024, accepted 12 August 2024, date of publication 27 August 2024, date of current version 10 September 2024. Digital Object Identifier 10.1109/ACCESS.2024.3450439

SURVEY

Human–Machine Interfaces: A Review for Autonomous Electric Vehicles

JESUS A. MANDUJANO-GRANILLO[®], MILTON O. CANDELA-LEAL, JUAN J. ORTIZ-VAZQUEZ[®], MAURICIO A. RAMIREZ-MORENO[®], (Member, IEEE), JUAN C. TUDON-MARTINEZ[®], (Member, IEEE), LUIS C. FELIX-HERRAN[®], ALFREDO GALVAN-GALVAN, AND JORGE DE J. LOZOYA-SANTOS

School of Engineering and Sciences, Tecnológico de Monterrey, Monterrey 64700, Mexico

Corresponding author: Juan C. Tudon-Martinez (jc.tudon@tec.mx)

This work was supported in part by Tecnológico de Monterrey and in part by CONAHCYT.

ABSTRACT Human-Machine Interfaces (HMIs) play an important role in ensuring performance reliability to create safety and comfort during the operation of autonomous vehicles, by efficiently communicating their current state and driving intentions to drivers, passengers, and pedestrians. The motivation for this study emerges from the increasing development of autonomous vehicles and the critical need for effective HMIs to bridge the communication gap between these vehicles and their users. Examples of HMI technologies include visual displays for status updates, auditory alerts for immediate hazards, and haptic feedback for enhanced control. During automated driving tasks that may require users to take control of the vehicle under certain conditions, the interfaces must support the situational awareness of the drivers by providing effective distraction, and achieving seamless integration with advanced vehicle control systems. The objective of this literature review is to present a comprehensive overview of the development of human-machine interface technologies in autonomous vehicles in different domain areas and to identify design challenges and opportunities to improve their interaction with their users.

INDEX TERMS Human-machine interface, autonomous vehicles, automated driving systems, performance reliability, situational awareness, user experience, vehicle-to-pedestrian communication, biometrics.

I. INTRODUCTION

Autonomous Vehicles (AVs) are a constantly evolving platform, their advances in technology have been implemented in commercial vehicles throughout the years. These advances range from the development of cruise control and anti-lock brakes to the introduction of advanced driver assistance features such as automatic emergency braking, which increases user safety while driving [1]. Autonomous vehicles are capable of performing automated driving tasks with the combination of sensors for environment perception, and navigation algorithms, but there are challenges in AV design regarding reliability to safety, comfort, and improving the situational awareness of the driver [2].

One important aspect of AV design is the development of effective Human-Machine Interfaces (HMIs) that provide

The associate editor coordinating the review of this manuscript and approving it for publication was Giuseppe Desolda^(D).

a complete and intuitive interaction between the automated vehicle and its users [3]. Developments of HMIs in the automotive industry began with basic mechanical displays and simple infotainment systems. Innovations like CAN technology, Bluetooth connectivity, and touchscreens have led to the integration of advanced driver assistance systems and automated driving features, which have changed the focus of modern HMIs to enhance safety and user experience. HMIs in autonomous vehicles are essential to communicate the current state of the vehicle, and subsequent actions/intentions to drivers, passengers, and pedestrians, as well as being capable of receiving feedback from them to make decisions. Effective HMIs are critical for bridging the communication gap between the vehicle and its users, ensuring that drivers remain informed and can take control when necessary. Examples of HMI technologies include visual displays for presenting critical information, auditory alerts to warn of immediate hazards, and haptic feedback to provide physical

sensations that enhance the driver's control and awareness. Current AV technology is only capable of operating at an automation level that requires the users to be alert of their surroundings while being prepared to take control of the vehicle under certain conditions, however, an ideal design of the HMI should be able to support the driver's situational awareness during non-driving related tasks (NDRTs) such as interacting with other passengers, using infotainment systems, using a personal device, among others [4]. The integration of advanced HMI technologies not only enhances safety by keeping drivers engaged and informed but also improves comfort and user experience by allowing seamless interaction with the vehicle's automated systems. Challenges in HMI design include creating interfaces that are intuitive and easy to use, minimizing driver distraction, and ensuring that the information provided is accurate and timely. These challenges are compounded by the need to design HMIs that can adapt to different levels of automation and varying user needs.

For this work, a systematic literature review of HMIs in autonomous vehicles is presented, where a structured process for search and selection of articles, as well as a synthesis and categorization of selected studies based on the systematic review protocol PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses) [5]. The objective of this systematic approach is to provide a comprehensive overview of the current state and challenges of HMI technologies for the interaction with users of autonomous vehicles, including drivers, passengers, and pedestrians. Detailed information on the study selection criteria and processes is provided in the Methodology section. The selection criteria for the review are shown in Figure 1.



FIGURE 1. Selection criteria for publication analysis.

A. RESEARCH QUESTIONS

According to the objective of this review, the following research questions were posed:

- Research Question 1: What is the state of the art in HMI technologies in autonomous vehicle applications, and what are the implications of these technologies for future research and implementation?
- Research Question 2: What are the trends among researchers in the field of HMI design and the experimental platforms that are being used to evaluate these technologies?
- Research Question 3: What are the challenges researchers face when designing and testing HMI technologies, and how are they addressed to ensure their proposals provide useful information about their performance?

B. CONTRIBUTIONS

The main contributions that this review provides are:

- A broad understanding of developments in Human-Machine Interface technology: By analyzing the historical and current developments of HMI technologies, this review aims to offer a clear perspective of their evolution, and how this is supporting technology development in the field of autonomous vehicles.
- An in-depth insight into emerging trends and novel technological developments in HMIs for AVs: This review presents the latest developments in HMI technologies focused on AV applications, detailing common experimental platforms and performance indexes used for evaluating design concepts.
- An evaluation of the challenges and limitations of HMI research strategies: Finally, this work analyzes the challenges and limitations faced by researchers when developing and evaluating the performance of HMI technologies.

This review offers a comprehensive and up-to-date examination of HMI technology developments, including historical background, emerging trends, and evaluations of research strategies, with a particular emphasis on autonomous vehicle experimental platforms, applications, and performance assessment methodologies. In contrast to other reviews on the subject, such as [6], which focuses solely on external HMIs, and [7], which covers HMIs across different applications, this review categorizes HMIs based on the user interaction, whether by passengers inside the vehicle or pedestrians outside the vehicle.

This article is presented in the following structure. Section 2 provides a conceptual background on the key concepts and challenges regarding autonomous vehicles and HMIs. Section 3 describes the systematic approach that was carried out for this review, detailing the steps and search, the selection criteria, goals, classification, and literature complexity, while identifying a common architecture among authors regarding HMIs for autonomous vehicles. Section 4 presents the historical developments on HMIs, as well as the design and structure used for autonomous vehicle applications whereas Section 5 presents different aspects of HMIs for AVs detailing each experimental platform, commercial technology, trends, and challenges. A discussion of the observed challenges of current HMI developments is established in Section 6. Finally, the conclusions of the findings are presented in Section 7.

II. SCOPE AND CONCEPTUAL FRAMEWORK

To provide context to the analysis, we begin by presenting the challenges that are present in electric AVs, which is the main research subject of the study. Then, additional relevant concepts are described such as the different autonomy levels that AVs can operate, the different vehicle classifications according to their size, the different architectures of HMIs in vehicles, and the use of biometrics of drivers, passengers, and pedestrians.

A. ELECTRIC AUTONOMOUS VEHICLE CHALLENGES

Electric autonomous vehicles have been regarded as the primary solution to many environmental and social issues in the automotive sector. These vehicles are equipped with multiple advanced technologies that allow them to recognize the environment they are operating in, and make decisions regarding the necessary action to ensure reliability for a proper driving experience. The most common sensors in AVs, as displayed in Figure 2 are as follows:

- Light Detection and Ranging (LiDAR): Provides high-resolution 3D mapping of the vehicle's surround-ings, aiding in object detection, localization, and navigation [8].
- **Cameras:** Used for visual perception, object detection, traffic signage recognition, lane detection, and pedestrian detection [9].
- **Radar:** Offers long-range detection capabilities, especially in adverse weather conditions, for detecting vehicles, pedestrians, and obstacles [10].
- Ultrasonic Sensors: Assist in close-range object detection, particularly useful for parking assistance and maneuvering in tight spaces [11].
- Global Positioning System (GPS): Provides precise localization data, essential for navigation and route planning [12].
- Inertial Measurement Unit (IMU): Measures vehicle motion and orientation, aiding in localization, navigation, and stabilization [13].
- Environmental: Including temperature, humidity, and barometric pressure sensors, to gather data about the vehicle's surroundings for enhanced situational awareness [14].

The common AV system architecture, as displayed in Figure 4, consists of a perception layer that acquires environmental information through the use of sensors and the fusion and processing of collected data; such data is then analyzed and interpreted by a decision layer which generates planning information; the planning information is sent to an action layer that takes control of the vehicle according to commands from the decision layer. Recent advancements have focused on improving the decision-making and control



(a) Autonomous vehicle external view



(b) Autonomous vehicle internal view

FIGURE 2. Schematic of common sensors and actuators in an autonomous vehicle (a) exterior view, (b) interior view.

processes within this architecture. For example, neural network-based model reference adaptive control (MRAC), have been explored to enhance the robustness and accuracy of the decision-making process in dynamic environments [15]. Additionally, model predictive control techniques can be applied to optimize the performance of electric motors and other critical components in real-time, ensuring efficient and reliable operation [16]. Furthermore, robust control methods, such as Particle Swarm Optimization (PSO) tuned controllers, have been developed for UAVs to stabilize altitude and track trajectories effectively, showcasing the potential for application in AVs to improve navigation and stability [17]. Although HMIs in AVs don't provide explicit information about control strategies, they focus on conveying insights into the environmental model and action strategies. One application of these technologies in a commercial AV setting can be seen with Waymo OneTM ride-hailing service which utilizes an HMI design shown in Figure 3 that displays the vehicle's current route, LiDAR detection of nearby vehicles, and traffic light statuses at intersections which ensures clear communication with the user and can be personalized to their preferences such as language, enhancing safety by providing transparent, realtime insights into the autonomous vehicle's perception and decision-making processes.



FIGURE 3. Waymo OneTM HMI example with environmental model elements.

There are still important political, ethical, safety, and legal issues that limit the deployment and reliability of autonomous vehicles [19]. Reliability assessment of AV systems is in high demand before they are launched on the market, the concept of reliability must be expanded to consider more safety-related issues [20]. One way to address reliability is through HMI design; elements of each layer of the AV need to work in coordination to minimize errors that could negatively impact the driving experience and the information of these elements can be presented through the use of HMIs so that the AV users can be informed about the vehicle status, and be alerted if there are conditions that require them to take control of the vehicle. For instance, genetic algorithm-tuned controllers have shown promise in optimizing control systems to handle uncertainties and improve overall performance reliability [21]. Additionally, solving optimal path planning problems using advanced techniques like the Bidirectional Rapidly-exploring Random Tree Star (BRRT*) combined with the Dynamic Window Approach (DWA) and Adaptive Monte Carlo Localization (AMCL) has been crucial for ensuring precise and reliable navigation in dynamic environments [22]. The challenges of HMI design involve improving users' perceived safety, reliability, and experience to increase the acceptance of autonomous vehicles [18]. Additionally, because the high complexity and dimensionality of driving behavior, the efficient evaluation of the multidimensional performance of autonomous vehicles in simulation and real-world scenarios is a long-established problem [23].

B. AUTOMATION LEVELS AND VEHICLE CLASSIFICATION BY SIZE

The automated driving characteristics of a vehicle are commonly defined in function to the required human intervention needed to operate. The Society of Automotive Engineers (SAE) has proposed six different levels of automation, separated into driver support features for the first three levels and automated driving features for the remaining levels [24]. In addition to the SAE levels of automation, vehicle classification by size is another relevant concept for automotive performance assessments. According to different national automotive groups and associations such as the National Highway Traffic Safety Administration (NHSTA) and the Automotive Science Group (ASG), vehicles can be classified by their size, separating into different classes in accordance to parameters such as curb weight, interior volume index, and gross vehicle weight rating [25], [26].

C. HUMAN-MACHINE INTERFACES IN VEHICLES

Automotive Human-Machine Interface (HMI) refers to the interface between human and machine control systems, including software, hardware, interfaces such as dashboard displays, infotainment systems, touch screens, voice recognition systems, and other control mechanisms used to interact with the vehicle's onboard systems and functions. In autonomous vehicles, the HMI includes interfaces designed to facilitate user interaction related to control with the vehicle's autonomous driving functions, such as setting navigation destinations, monitoring system status, and overriding automated controls when necessary. HMI functionalities differ depending on the automation level of the vehicle they are implemented. Lower levels may only focus on displaying the current state of the vehicle characteristics to the user, while higher levels should inform additional information regarding the driving mode of the vehicle, environmental information, and even motion and maneuver intentions to provide awareness of the surroundings [27].

D. BIOMETRICS OF DRIVERS, PASSENGERS AND PEDESTRIANS

Biometrics consists of the capturing of relevant physiological data and behavioral characteristics through sensor technology. These characteristics are commonly used for the development of HMI systems that can evaluate a driver's condition during driving tasks, such as drowsiness and vigilance alertness, via face-monitoring systems that measure changes in eye regions and particular head rotations that may indicate that the driver is distracted or fatigued [28]. Other biometric parameters that can be used to enhance HMI functionality include Blood Pressure-Volume (BVP), Galvanic Skin Response (GSR), Heart-Rate Variability (HRV), respiratory frequency, and hand trembling, among others. The general flow process of a biometric system, as presented in Figure 5, consists of an initial registration process, where biometric data is captured from individuals through a sensor unit; a feature extraction unit processes the sensed data to extract relevant information and store it in a local database unit. Next, in a separate recognition process, a biometric input is captured by the same sensor unit, where the feature extraction unit processes the data to extract the target characteristics, and then, through a matcher unit, the extracted features are compared with those from the database unit to feed the decision algorithm of the application they are implemented in [7].

IEEE Access



FIGURE 4. Architecture of an autonomous vehicle, image modified from Wang et al [18].





FIGURE 5. Process flow of a biometric system.

III. METHODOLOGY

This section details the process that was carried out to conduct the systematic literature review. The gathered literature was selected based on specific search methods and criteria that ensured an updated state-of-the-art revision of HMI technologies in autonomous vehicles. The review followed the systematic review protocol PRISMA to ensure transparency and reproducibility. Six stages have been considered in this systematic literature review, briefly described in the next paragraphs.

A. STEPS AND SEARCH

In the identification stage, a comprehensive search of various types of publications including journal articles, conferences, and reviews, among others, was conducted using SCOPUS as the main research database. Specific keywords including "autonomous vehicles", "self-driving vehicles", "humanmachine interfaces", and "HMI", were used to ensure the study's themes were related to the research topic.

B. SELECTION

Following identification, the screening process involved reviewing the titles and abstracts to exclude duplicates and irrelevant studies. The screening process ensured that the remaining studies met the inclusion criteria by focusing on their relevance to the topic and their contribution to understanding HMI technologies in autonomous vehicles.

For determining eligibility, an assessment of the full texts of the screened articles was conducted. The criteria for eligibility included relevance to the development and application of HMI technologies in autonomous vehicles, a focus on interaction with users (including drivers, passengers, and pedestrians), studies published between 2010 and the present to ensure the inclusion of recent advancements and trends, and the inclusion of empirical data, experimental methods, or comprehensive reviews related to HMIs in autonomous vehicles.

Once the set of keywords and the research database were defined for the search process, the following selection criteria was established, as Figure 1 illustrates:

- Year of publication: 2010 to present.
- Publications with the terms "autonomous vehicles" (or similar), "HMI", "interfaces" in their keywords or title.
- Selection priority to publications by number of citations in SCOPUS database search.
- Publications whose HMI study subject presented a SAE Level of automation between 2 and 4.

Studies with descriptive elements of HMIs in the automotive sector with an emphasis on autonomous vehicles were the main objective of the selection process.

This structured process ensured that the review was comprehensive and included high-quality studies that provide valuable insights into the state and challenges of HMI technologies in autonomous vehicles.

C. ARCHITECTURE OF HMIs IN AUTONOMOUS VEHICLES

The selected papers were classified according to the HMI architecture used in cars, specifically focusing on internal and external interfaces.

The general architecture of HMIs in autonomous vehicles consists of an input channel and an output channel. The input channel can perceive inputs not solely from the drivers but also from passengers and external factors such as pedestrians, weather and road conditions, and other vehicles through the use of sensors, which can include visual technology, biometrics, etc. The output channel consists of extracting features from the input channel and communicating them internally to the users of the automated vehicle and externally to pedestrians. To better understand the specific roles and design challenges of HMIs, it is useful to distinguish them between internal and external interfaces. Internal interfaces focus on the interaction between the vehicle and its users, primarily the driver and passengers. These interfaces are responsible for providing information about the vehicle's status, upcoming maneuvers, and situational awareness. Effective internal HMIs ensure that drivers remain informed and engaged, reducing the risk of accidents and improving overall safety. They also enhance the user experience by integrating infotainment systems, navigation aids, and personalized settings.

External interfaces, on the other hand, are designed to communicate the vehicle's intentions to external entities such as pedestrians. These interfaces play a crucial role in ensuring the safety of vulnerable road users by providing clear and intuitive signals about the vehicle's actions, such as stopping, turning, or yielding. External HMIs often use visual displays, lights, and sounds to communicate messages effectively. The distinction between internal and external interfaces in autonomous vehicles is critical because it highlights the different user groups and interaction contexts that each type of interface must address.

This architecture framework was chosen because it provides a comprehensive approach to understanding the multiple interactions that are present in autonomous vehicles. By categorizing HMIs into internal and external interfaces, we can more effectively analyze and address the specific design challenges and opportunities associated with each type. This classification aligns with the main goals of this review, which are to present a comprehensive overview of HMI technologies, identify design challenges, and explore opportunities to improve user interaction with autonomous vehicles.

Figure 6 presents an architecture of the external and internal interactions performed in autonomous vehicles based on [27] to illustrate the feedback loops between the vehicle and its various user groups.

For internal interactions, the HMI input from the driver and passengers to the AV includes functions such as navigation, driving mode selection, communication, control override, customization settings, and comfort controls. These inputs help in guiding the AV's actions and ensuring that the vehicle operates according to the user's preferences and needs. On the other hand, the HMI feedback from the AV to the driver and passengers includes critical information about vehicle condition, environmental conditions, system alerts, automation status, motion intentions, and energy consumption. This feedback loop ensures that users are kept informed about the vehicle's status and can make informed decisions or take control when necessary.

For external interactions, the architecture encompasses interactions with the external environment, pedestrians, and road users. The AV receives environmental data such as road type, traffic rules, weather conditions, and infrastructure information, which are crucial for safe navigation and compliance with traffic laws. In interactions with pedestrians, the AV detects pedestrian presence and location, gestures, and crossing intentions through sensors. It communicates



FIGURE 6. General architecture of interactions in autonomous vehicles.

back to pedestrians with information about the vehicle's current status, motion intentions, and operational status, enhancing pedestrian safety and confidence in interacting with autonomous vehicles. For road users, the AV gathers data on the number of vehicles, their speed, types, and performs motion analysis to navigate effectively and safely in traffic.

D. GOALS

The goal of this literature review is to present trends in the development of HMI technologies in autonomous vehicles in different areas of the previously stated architecture, and the challenges it entails.

E. CLASSIFICATION

The selected studies were sub-categorized in accordance to the following topics:

- 1) History of HMIs in autonomous vehicles.
- 2) Designs and experimental developments in HMIs for autonomous vehicles.
- 3) Biometrics in autonomous vehicles' HMIs.
- Specific application of autonomous vehicles and their HMIs.

F. COMPLEXITY

The complexity of each topic is intended to present the main contributors to the development of the different HMI technologies in autonomous vehicles, as well as expanding

into existing standards and regulations, and describing trends and challenges.

IV. LITERATURE REVIEW ON HMIS: HISTORICAL DEVELOPMENTS, DESIGN AND STRUCTURE FOR AUTONOMOUS VEHICLES

In recent years, research in the field of HMIs in autonomous vehicles has seen quick growth. Analyzing the scientometric information of the literature review, as shown in Figure 7, using the keywords combination "autonomous vehicles" AND "human-machine interface" AND "systems", it is shown that research in this topic began to gain momentum around 2010, and it exponentially grew from 2016 onward, and it has continued to increase, demonstrating that the field has a lot of potential and research opportunities. The main contributors to research on the topic are Germany, the United Kingdom, and the United States, with each country making significant contributions. Other countries, such as China, Italy, and Japan, among others, have also shown developments in the area. Most of the research in the field of HMIs in autonomous vehicles has been published in the form of articles and conference papers, with some book chapters and reviews also available. The main subject areas involved in the research are engineering, computer science, social science, and mathematics.

In this section, we begin detailing the historical developments of HMI technologies in the automotive sector. Next, different uses of HMIs in autonomous vehicles will be presented, including standards and regulations, and expanding on the common approach that is being used for the development of internal and external human-machine interfaces.

A. HISTORY OF HMIs IN VEHICLES

Most of the current research in the field of HMIs in autonomous vehicles focuses on the development of interfaces that enhance performance reliability to improve the safety and experience of the users. However, it is important to analyze the historic developments of interfaces in the automotive industry since they reflect the changes in preferences and expectations of drivers and passengers that have impacted their design and evaluation methods.

Figure 8 shows a timeline of interface developments in vehicles. Meixner et al. [29] provide an insight into the evolution of HMIs in the automotive sector. HMI developments started with purely mechanical displays in 1915. Manufacturers moved displays to less distracting locations and introduced entertainment features like car radios, in-vehicle phones, and cassette and CD players. Bosch introduced CAN technology in 1986, allowing communication between devices and the development of infotainment systems that, like Mercedez-Benz CNS in 1993, led to models and guidelines being introduced regarding safe HMI design. By this time, developments in Advanced Driver Assistance systems (ADAS) began to emerge. In the 1980s, driver assistance systems used proprioceptive sensors that measured the status of the vehicle and allowed controlling the vehicle dynamics, which then evolved to the use of exteroceptive sensors which measured external conditions of the vehicle to provide information and warnings [30]. Technological innovations like Bluetooth and internet connectivity were implemented in the 2000s, along with resistive touchpads. Additional developments in automotive HMIs involve the transition from resistive to capacitive touchpads for improved response, and the introduction of combiner and windshield displays that decreased eyestrain and presented driving information in the driver's field of view [31]. Finally, in the 2010s, further developments for driver assistance systems that provided automated driving features were accomplished.

B. HUMAN–MACHINE INTERFACE DESIGN IN AUTONOMOUS VEHICLES

A common approach for the design of HMIs in autonomous vehicles relates to the emphasis on ensuring the safety of the user through the use of different stimuli such as visual, auditory, and haptic that assist and encourage the interaction with the interface [32]. Additionally, it is important to make a distinction between interfaces that interact with both drivers and passengers, which are commonly referred to as "internal HMIs" (iHMI), and vehicle interfaces that communicate with pedestrians denoted as "external HMIs" (eHMI). In both cases, HMI design should be usercentered, which involves providing transparency about the autonomous system by presenting information in a simple,





(d) Number of publications by country

FIGURE 7. Scientometrics of SCOPUS search by (a) year, (b) type of publication, (c) subject area, and (d) country.

but understandable manner to avoid overloading the users with non-relevant information that could negatively impact



FIGURE 8. Timeline of developments of HMIs.

their decision-making under manual and autonomous driving circumstances, which would also decrease the system's reliability and affect the acceptance of automated vehicles [33]. Figure 9 shows the distinction between the different types of human-machine interfaces in autonomous vehicles, detailing the technologies, simulation, and experimental platforms and type of information that have been documented.

1) INTERNAL HMIs IN AUTONOMOUS VEHICLES

Internal HMIs of autonomous vehicles focus on providing information about the vehicle and its surroundings to both drivers and passengers. Research regarding this topic has been primarily focused on the elements and characteristics of the interface that improve safety, comfort, user experience, and acceptance of autonomous vehicles. One of the most important aspects of HMI design for safety is improving the situational awareness of drivers in an environment where autonomous driving is being performed while drivers perform non-driving related tasks [45]. An example of the HMI elements that can improve this situational awareness is the implementation of displays that present information about the status of the vehicle between the drivers and passengers in a design concept that allows the driver to perform other activities that also improve their comfort [46]. Another factor that can contribute to improving the situational awareness of a driver who is not performing driving activities is the addition of auditory stimulus that decreases the visual workload required to monitor the vehicle information through the use of only visual displays [47].

Multiple studies have corroborated that one common performance index that can be tested to indicate safety in HMI design is the take-over time, which involves the time drivers take from when the HMI in the vehicle indicates to them that manual driving is required, to when they regain control of the vehicle [48]. Other alternative actions to increase the situational awareness of users via HMI design can be through the use of different stimuli such as visual, auditory, and vibrotactile signals that can impact the drivers' responses, with a combination of these, take-over processes can be done more quickly to avoid any negative impact on the driver's manual driving behavior [49], which is essential for ensuring performance reliability and experience in an automated vehicle.

Other aspects that have been studied are related to how the HMI elements must be presented to improve the users' understanding of the system alerts and information on driving conditions. Figure 10 shows an example of an internal HMI concept that communicates many aspects of an autonomous vehicle. Elements such as color coding and the use of standardized and recognizable signs for announcing relevant events can improve how drivers interpret the messages of HMIs in autonomous vehicles [50], [51]. The use of augmented reality (AR) can be implemented in conjunction with the visual technologies implemented in the perception layer of the autonomous vehicles to display information about the road, environmental conditions, and infrastructure, among others, HMI design can take advantage of this by indicating the users the identified road and using graphic elements such as carpet and arrows over the displayed road to indicate maneuver intentions of the autonomous vehicle such as lane changes, braking, speed regulation based on road signals, etc [52], [53].

2) EXTERNAL HMIs IN AUTONOMOUS VEHICLES

External HMIs' primary function relies on transmitting the current vehicle's state and its intent information for short future to the rest of the traffic participants [6]. Figure 11 shows the interaction between vehicles and pedestrians using vehicle sensors to capture human features and display them through vehicle actuators for communication. The topic of research for external HMIs focuses on the development of improving communications using V2X devices between the autonomous vehicle and traffic participants such as traffic signage infrastructure, pedestrians and other vehicles driven by humans with a specific focus on the interaction with



FIGURE 9. Human-machine interfaces in AVs.



FIGURE 10. Internal visual HMI design concept for communicating vehicle status, motion intention and maneuvers.

pedestrians since they are the most vulnerable group in human-machine interactions.

In conventional vehicles, the use of lights and headlights makes communication possible in everyday interactions and situations including challenging environments where reduced visibility is present. In autonomous vehicles, external HMIs are used to accelerate pedestrians' inferences about their motion intentions [54], [55]. The visual design of autonomous vehicles is an aspect that makes them stand out from commercial cars. Perception technologies such as LiDARs and cameras may indicate to pedestrians that a different type of vehicle is in front of them, so additional

visual elements are needed to indicate driving conditions and intentions, such as visual displays in combination with LED strips that communicate with pedestrians via text and symbols as demonstrated in Figure 12. The use of these elements can improve decision-making in crossing situations and increase the safety perception of pedestrians [56], [57]. In external HMI design, there are certain characteristics for the implementation of the previously described visual elements such as color, positioning, and contents that can affect communication with pedestrians. External HMIs positioned on the roof of the autonomous vehicle that transmit messages to pedestrians in crosswalking situations indicating actions that they are allowed to perform safely like "walk", have had a higher preference for perceived safety over preventive warning messages that explicitly state actions that pedestrians should not do like "don't walk" [58].

An additional characteristic of human-machine interactions relies on the use of auditory stimuli for communication. Sound as a method of communication is something relatable to pedestrians since conventional vehicles use speakers to indicate warnings and engine sounds to indicate their presence. For autonomous vehicles, their electric nature makes their engines produce different sounds with typically lower intensities, so alerting sounds and soft warnings are methods by which they can communicate with pedestrians without the need to look at the vehicle, such as the use of artificial sound generators, in combination with visual elements, can increase performance reliability and safety [60]. Pedestrians may also



FIGURE 11. eHMI interaction scheme.



FIGURE 12. External HMI visual elements proposals: a) LED Bar, b) Bumper Panel, c) Windshield, d) Hood, e) Road Projections, f) Vehicle Mask, g) Interior Panel, image modified from Orlický et al. [59].

prefer that the messages be displayed visually in the external HMI, and AVs should also be communicated through audio due to environmental conditions that could affect its visibility such as direct sunlight or foggy conditions, or physiological limitations such as visual impairments [61].

A seemingly overlooked aspect in external HMIs is related to the cultural factors that can affect the effectiveness of their design. Certain characteristics of the external HMI such as symbols, and colors may convey different messages depending on the country, and pedestrian behavior, cultural norms, and expectations about technology and automation could impact their reactions to automated vehicles [62]. To this extent, cultural adjustments need to be considered upon the decision to implement this type of technology.

C. STANDARDS AND REGULATIONS FOR AUTONOMOUS VEHICLES HMIs

Currently, as far as the authors have researched, there are no official standards that focus on guidelines for the design of both types of HMIs in autonomous vehicles. The NHTSA in the United States does not have specific performance reliability requirements to design for HMI systems in autonomous vehicles. However, NHTSA's regulations and guidelines aim to ensure the overall safety and effectiveness of vehicle operation, including aspects related to HMI performance. Specifically, NHTSA's Federal Motor Vehicle

Safety Standards (FMVSS) cover various aspects of vehicle design, construction, and performance, which indirectly influence HMI reliability. For example, FMVSS 101 specifies requirements for controls and displays, including factors such as visibility, readability, and ease of use, which are critical for HMI performance [34]; while FMVSS 135 addresses brake systems, including warning light indicators [35]. FMVSS 126 addresses electronic stability control (ESC) systems, including HMI interfaces, to enhance vehicle stability and prevent rollover accidents, and others [36]. While NHTSA has not issued specific reliability requirements for HMI in autonomous vehicles, automakers, and technology developers typically adhere to industry best practices and standards to ensure the reliability, usability, and safety of HMI systems. This may involve rigorous testing, validation, and human factors engineering to address potential failure modes, usability issues, and safety concerns related to HMI design and functionality. Additionally, companies often conduct user studies and gather feedback to iteratively improve HMI reliability and user experience over time.

Other existing standards related to HMI design in autonomous vehicles come from the International Organization for Standardization (ISO). For instance, ISO 26262 specifies functional safety requirements for automotive systems, including HMI components, to mitigate risks associated with malfunctions and failures [37]. ISO 15005 focuses on the ergonomics of human-system interaction in vehicles, addressing HMI design principles, display legibility, and user interface considerations [38]. ISO 9241 covers humancentered design principles for interactive systems, including HMI interfaces, emphasizing usability, accessibility, and user experience [39].

On the other hand, the SAE has proposed the following standards: SAE J3016 defines levels of driving automation and terminology for autonomous vehicle systems, including HMI functionalities [24], and SAE J2395 specifies guidelines for vehicle user interfaces to ensure safe and intuitive interaction with automated driving systems [40].



FIGURE 13. Guidelines for HMI design in autonomous vehicles, image based on Naujoks et al [44].

Despite the lack of specific regulation for HMIs in autonomous vehicles, there have been proposed guidelines and models that cover many aspects of the topic, such as design for comfort, wellness, and experience, among others, that are based on the demands of different previously stated regulatory bodies and associations or existing design models such as Benyon's "People-Activity-Context-Technology" (PACT) framework [41], [42]. One example of these proposed guidelines is shown in Figure 13; the contents of the guideline establish principles, recommendations, standards, criteria, and verification methods to guide the development of HMIs in autonomous vehicles. These guidelines have been evaluated by a workshop on evaluation methods for autonomous vehicles [43], and reflect multiple aspects that have been experimented with for the design of HMIs in autonomous vehicles.

V. LITERATURE REVIEW ON HMIS FOR AUTONOMOUS VEHICLES: EXPERIMENTAL PLATFORMS, TECHNOLOGIES AND PROGRESS BY DOMAIN AREA

In this section, interesting experimental vehicle platforms, using internal and external HMIs, are presented for different novel actuation systems used for HMIs are detailed. Then, an overview of the use of biometrics technologies for HMIs is presented, analyzing the main interaction principles that are commonly developed for driver assistance systems. Finally, a review of different current specific applications of HMI technology in autonomous vehicles is demonstrated.

A. EXPERIMENTAL PLATFORMS FOR HMI DESIGN IN AUTONOMOUS VEHICLES

purposes identifying their characteristics and performance

indexes. Technologies based on Artificial intelligence and

Simulation and experimental platforms have been developed to test and evaluate design concept performance reliability for both internal and external HMIs. For internal HMIs, as shown in Figure 14, they range from the use of simple driving simulators, where a monitor is used for studying the driving style of a driver, to cabin simulators that emulate the interior of an autonomous vehicle. In a cabin simulator, the point of view is shown in front of the cabin's windshield and different sensors inside the cabin are used to capture the passengers' perception. During these simulations users are typically presented in situations where automated driving is present and an external condition is introduced for the HMI to inform the user that attention is required, then the elements of the interface are presented either via the simulation or physically at the user's side to evaluate how the experimental users react and regain control of the vehicle [63]. Some of the parameters that are tested for internal HMI experiments include reaction times and take-over times, sometimes including additional details regarding the actions users do after taking control of the vehicle such as speed changes or gaze positions while driving [64]. Additionally, authors may also gather subjective information about the experiment subjects' perceived safety, performance reliability, and acceptance of the experimental platform through the use of evaluation ratings that can be used to enhance the platform by changing, adding, or subtracting elements based on users feedback [65], [66].



FIGURE 14. Simulation cabin setup "COMPACT simulator" manufactured by VI-grade for evaluating internal vehicle HMI [67].

Experiments for external HMIs (eHMIs) are similar in the use of simulation platforms. In this case, there is a prevalent use of simple computer simulations to simulate pedestrians in crossing situations, however, there are cases where the use of technologies such as virtual reality can enhance the simulation to produce more reliable results [59]. However, in contrast to experiments with internal HMI designs, as seen in Figure 15, external HMI experiments have had on-field tests with functional physical eHMI concepts for autonomous vehicles to evaluate their effectiveness under certain experimental conditions, primarily perceived safety and performance reliability from pedestrians [68], [69].

Table 1 shows different experiments presented in literature regarding internal and external HMIs, specifying the type of HMI that was experimented on, the experimental platform that was used for testing, and the characteristics of the HMI that were evaluated with their respective performance indexes. It can be observed that internal HMI experiments tend to solely use simulations to test design concepts, testing graphical elements displayed in the HMI with some experiments analyzing the use of auditory feedback. In the case of external HMIs, there is a mixture of both simulation and on-field testing platforms to evaluate design elements. Specifically, there is a focus on analyzing the effectiveness of



FIGURE 15. On-field experimental autonomous vehicle platform to evaluate HMI elements at Tecnológico de Monterrey.

visual elements for eHMIs to ensure performance reliability in pedestrians during crossing situations.

Table 1 can be used to compare the recent experimental vehicle platforms with internal or external HMIs, by identifying the performance indexes used to determine the evaluation criteria proper for both HMI design concepts. Evaluating the effectiveness of HMIs in autonomous vehicles requires robust methodologies and standardized frameworks to ensure reliability and validity. Both types of HMIs are predominantly evaluated focusing on metrics such as interaction duration, perceived performance reliability, and ease of use and understanding. These methods allow for controlled testing conditions but may not fully capture the complexities of real-world interactions.

In addition to minimizing driver distraction and ensuring intuitive interaction, it is essential to consider how endusers perceive and interact with HMI systems in autonomous vehicles. Various studies have utilized user rating evaluations to assess the usability, attractiveness, and effectiveness of HMI designs. For instance, research has employed online surveys with industry experts to evaluate multimodal HMI frameworks by measuring hedonic and pragmatic qualities. Other studies have implemented simulation environments to evaluate system usability, understandability, and potential overreliance on automation, employing scales for task load assessment. These evaluations provide insights into user preferences, highlighting the importance of designing HMIs that are not only functional but also align with user expectations and comfort levels.

An opportunity area for both types of HMI research is the lack of emphasis on analyzing the effects of auditory stimulus for safety and performance reliability perception. For internal

TABLE 1. HMI simulation and experiments categorization.

Study	Year	НМІ Туре	Experimental Platform	Tested characteristic	Performance indexes
[63]	2015	Internal	Computer simulation	Graphical elements	Take over time
[66]	2016	Internal	Computer simulation	Graphical and interactive elements	Interaction duration and users' perceived performance reliability, performance, ease of use and stressfulness
[70]	2016	Internal	Simulation cabin	Graphical elements	Users' perceived usefulness and satisfaction
[51]	2016	Internal	Computer simulation	Graphical elements	Users' perceived usefulness, ease of use and comprehension
[50]	2017	Internal	Computer simulation	Graphical elements	Comprehensibility ratings
[47]	2017	Internal	Simulation cabin	Graphical elements and auditory output	Monitoring ratio, perceived usefulness and perceived visual workload
[71]	2018	Internal	Simulation cabin and computer simulation	Graphical elements	User experience, reaction time, number of collisions
[52]	2018	Internal	Simulation cabin	Augmented reality visual elements	Take over time, eyes on windshield time, brake reaction time, lane change time, head angle
[64]	2019	Internal	Simulation cabin and computer simulation	Graphical and auditory elements	Average trajectories after take over request, and reaction time
[48]	2020	Internal	Computer simulation	Graphical elements	Take-over times
[33]	2021	Internal	Computer simulation	Graphical elements	User experience rating
[65]	2021	Internal	Simulation cabin	Graphical elements	Users' perceived performance reliability
[46]	2021	Internal	Simulation cabin	Interior design concept	Take-over times and situational awareness ratings
[53]	2021	Internal	Computer simulation	Augmented reality visual elements	Users' state anxiety, situation awareness rating, acceptance, take-over times, perceived controllabillity, deceleration and lateral control after take-over
[49]	2022	Internal	Simulation cabin	Auditory and tactile elements	Take over time, reaction time
[42]	2022	Internal	Prototype concepts	Graphical and interactive elements	User experience rating
[45]	2022	internal	Computer simulation	Graphic elements	User experience, perception and comprehension
[69]	2019	External	On-field testing	External visual elements	Perceived safety, performance reliability, acceptance and understanding
[68]	2020	External	On-field testing	External visual elements	User experience and perceived safety
[55]	2021	External	Virtual reality lab and computer simulation	External visual elements	Pedestrian and gaze behaviour
[54]	2021	External	On-field testing	External visual elements	User experience questionnaire
[58]	2021	External	Computer simulation	External visual elements	User experience rating and response time
[57]	2021	External	Computer simulation	External visual elements	User questionnaire
[59]	2021	External	Computer simulations	External visual elements	Visibility density and efficiency
[60]	2021	External	On-field testing	External visual and auditory elements	Trust, perceived safety, pragmatic quality, social presence, and transparency
[56]	2022	External	Visual questionnaire	External visual elements	Users' preference rating
[61]	2022	External	On-field testing	External visual elements	User's perceived information clarity and safety

HMIs, the lack of on-field experimental platforms may be considered a limitation due to the requirement elevated costs of obtaining an autonomous vehicle for research purposes and the programming complexity it may involve, which in contrast with external HMI tests, typical experiments use conventional vehicle platforms, seemingly instrumented to be automated to convey an autonomous nature. These limitations may impact the results of the experiments since users are in a safe, constrained environment that is different from real life. The developments in autonomous vehicle technology could increase their access as experimental platforms for validation of HMI concepts that could improve drivers, passengers, and pedestrians' safety and acceptance.

B. ARTIFICIAL INTELLIGENCE TECHNOLOGIES IN HMI DESIGN FOR AUTONOMOUS VEHICLES

As the level of automation increases, Artificial Intelligence (AI) technologies, in the form of Machine Learning (ML), need to be implemented, as the HMI should accurately diagnose the situation in order to know if it's possible to switch between human and machine while driving the vehicle, as human override is still an option in high automation [72]. Some of the AI technologies used to reinforce the interaction between the user and the autonomous vehicle and their research interest in the last 5 years based on an additional SCOPUS search, shown in Figure 16, are the following:

- Natural Language Processing (NLP): AI will foster NLP integration allowing users to interact with the vehicle using voice commands, enabling hands-free operation for tasks such as navigation, entertainment control, and communication [73].
- **Predictive analytics:** AI methods can study user behavior, preferences, and contextual data to anticipate user needs and proactively suggest actions or adjust vehicle settings to improve comfort, convenience, and safety [74].
- **Situational awareness:** By integrating data from onboard sensors, cameras, and external sources, AI can provide contextual awareness to the vehicle interface, enabling adaptive responses to changing driving conditions, traffic situations, and user requirements [75].
- **Personalization:** AI-powered personalization features can tailor the vehicle interface to individual user preferences, including screen layouts, menu structures, and content recommendations, improving the overall user experience [76].
- **Cognitive load monitoring:** AI algorithms can monitor the cognitive load of the driver and passengers by analyzing factors such as gaze, facial expressions, and physiological signals, providing feedback or assistance to reduce distractions and ensure safe driving [77].
- **Predictive Maintenance:** AI-powered predictive maintenance systems can monitor vehicle performance data in real-time, detect potential problems before they escalate, and recommend preventative maintenance actions to optimize vehicle reliability and lifespan [78].



FIGURE 16. Research distribution of AI-powered technologies in AVs over the past five years, based on SCOPUS data. The number of related research projects is in parentheses.

Examples of complex systems include an ADAS, which may use a Robot Operating System (ROS) and a Car Learning to Act (CARLA) simulator to receive and process Application Programming Interface (API) data based on Photoplethysmography (PPG) and Galvanic Skin Response (GSR) biosignals for emotion recognition and so assess whether a driver is capable of performing manual driving [79].

AI technologies can also enhance the real-time adaptability and learning capabilities of HMIs, enabling more effective interaction between users and autonomous vehicles. Recent AI-driven breakthroughs have significantly improved the ability of HMIs to adapt to changing environments and user needs in real-time.

One example of such advancements is the application of deep learning models to enhance pedestrian detection systems in autonomous vehicles which can improve the accuracy of identifying pedestrians and predicting their movements, leading to safer and more responsive autonomous driving behaviors [80].

Another relevant advancement involves the use of deep learning and multimodal data fusion to improve the detection of emergency vehicles. This approach combines multiple data sources to provide timely and accurate alerts to drivers, enabling quick and appropriate responses in critical situations so that HMIs could have the potential enhance situational awareness and safety for both vehicle occupants and other road users, ensuring a more reliable interaction with the environment [81].

C. BIOMETRICS IN AUTONOMOUS VEHICLES HMIs

The integration of biometric sensors in HMIs is vital to implement supervision methodologies (camera vision, vehicle monitoring, and lane monitoring), and carry out permanent tests to evaluate the levels of attention, fatigue, and workload status of the drivers and/or passengers, as shown in Figure 17, on which complex algorithms, as well as intelligent processing, are both required in order to provide reliable autonomous functions [72]. When processing these biometric features with additional ML algorithms, such as Support Vector Machine (SVM) and AdaBoost, it is possible to detect cognitive load using gaze direction, head orientation, pupil area and diameter, and R-R interval (distance between heartbeats), in order to assess driver's state on having a conversation while driving, ordinary driving and doing arithmetic tasks [82].

Tests are currently being done on AV simulated environments, which include Virtual Reality (VR) porting and motion platform integration in order to apply tilt and pitch angles and thus provide, finally, predicting a *hazard index* based on GSR [83]. Furthermore, being able to affect perception time, response time, internal vehicle processing time, external transmission time; and although there have been many HMI modalities; the best performance and most usability are achieved when having a functional autonomous level 5 vehicle, as not only communication time is eliminated, but also driver time [84].

The most relevant biometric sensor technologies and their research interest in the last 5 years based on an additional SCOPUS search, shown in Figure 18, are categorized as follows:

- Facial Recognition: Utilized for driver authentication, monitoring driver attentiveness, detecting drowsiness, and providing personalized settings based on recognized individuals [85].
- **Eye-Tracking:** Monitors the driver's gaze direction, pupil dilation, and blink rate to assess attentiveness, fatigue, distraction, and intention to provide timely alerts or interventions [86].
- Heart Rate Monitoring: Measures the driver's heart rate to detect stress, fatigue, or medical emergencies, enabling appropriate responses and adjustments to driving conditions [87].
- Voice Recognition: Allows for hands-free interaction with the vehicle's controls, enabling voice commands for navigation, entertainment, climate control, and communication [88].
- Gesture Recognition: Enables intuitive control of HMI functions through hand movements or gestures, enhancing user experience and reducing cognitive load [89].
- Electroencephalography: Measures brain activity to assess cognitive workload, attention levels, and emotional states, providing insights into the driver's mental state for adaptive HMI responses [90].
- Skin Conductance Sensors: Detect changes in skin conductivity to gauge stress levels, emotional arousal, or discomfort, informing HMI adjustments and intervention strategies [91].

Literature shows that biometrics used in HMI application tests include PPG, which is an optical sensor placed within the driver's index finger and can detect blood pressure, volume change, and heart activity; as well as GSR, which is acquired to measure the skin conductance via either the middle and ring fingers or by sensors mounted within the steering wheel or the seat belt [92], thus being able to gather sweat secretion and body temperature, further predicting driver's emotions, workload state, and fitness level [79]. In addition, standard and thermal cameras can be used to gather vision data to indicate stress level, fitness level, emotional state, and driver intention [92]. On the other hand, more wearable and obtrusive devices include detecting brain waves based on Electroencephalography (EEG), which can predict stress and mental fatigue state [93]. An additional non-conventional unobtrusive sensor is pressure cells, which can detect whether the driver is in a non-optimal position by measuring nonuniform pressure distribution [94], also combined with video sensors [95].

Although respiration rate and HR were not useful in [96] for detecting concentration, it can identify abnormal conditions that would trigger a takeover scenario (such as drowsiness or shock), in which a level 3 highly automated vehicle would follow the complex takeover strategy described in Figure 19. The set of Emergency maneuvers (EM) and Minimum Risk Maneuver (MRM) is critical in order to avoid an accident, based on biometrics to determine whether the driver is capable of manual driving [96].

Interaction with the HMI requires input from hand and fingers based on sensors; this includes Infrared (IR) Systems, based on infrared sensors to track a position; Wearable Device systems, attached directly to them; Internal Measurement Unit (IMU) Systems; as well as Remote Control-Based Points, using existing remote control devices and pointers [97].

D. ACTUATOR TECHNOLOGIES IN HMIs FOR AUTONOMOUS VEHICLES

Regarding the representation of this data collected by biometrics, actuator technologies allow the system to present information or related alerts to the users through HMIs that contribute to creating immersive, intuitive, and userfriendly HMI experiences in autonomous vehicles, enhancing safety, comfort, and convenience for passengers and drivers alike [98]. These actuator technologies and their research interest in the last 5 years based on an additional SCOPUS search, shown in Figure 20, can be categorized as follows:

- **Displays:** Various types of displays, including LCD screens, OLED displays, Heads-Up Displays (HUDs), and Augmented Reality (AR) displays, provide visual feedback to the driver or passengers regarding navigation, vehicle status, surroundings, and entertainment [99].
- Haptic Feedback Systems: Incorporate tactile feedback mechanisms such as vibrating motors, actuators, or electrostatic forces to provide physical sensations to the user, enhancing the perception of touch and interaction with touchscreens, buttons, or controls [100].
- Auditory Feedback Systems: Utilize speakers and sound generation technologies to deliver auditory cues, alerts, notifications, and voice prompts to communicate information effectively to the driver or passengers [101].
- Steering Wheels and Pedals: Although traditional incar controls, these actuator technologies may incorporate haptic feedback or force-feedback mechanisms

IEEEAccess

Biometric Flow



FIGURE 17. Flow diagram of ADAS integrated with biometric devices.



FIGURE 18. Research distribution of biometric technologies in AVs over the past five years, based on SCOPUS data. The value in parentheses indicates the number of related research projects.



FIGURE 19. Flow diagram for a Level 3 automation decision-taking takeover strategy, involving driver availability detection through biometrics.

to enhance driver interaction and provide tactile cues during autonomous driving modes or manual interventions [102].

Biometric Technologies



- Touchscreens and Touch-sensitive Surfaces: Enable intuitive interaction with HMI controls and interfaces through touch gestures, multitouch capabilities, and customizable user interfaces for accessing vehicle functions, navigation, entertainment, and climate control [103].
- Ambient Lighting Systems: Utilize LED lighting elements strategically placed within the vehicle's interior to provide visual cues, notifications, mood lighting, and personalized ambiance, enhancing user experience and comfort [104].
- Dynamic Seating and Interior Configurations: Employ actuators and motors to adjust seating positions, seat configurations, and interior layouts dynamically, optimizing comfort, accessibility, and space utilization based on user preferences, driving modes, or autonomous driving scenarios [105].



FIGURE 20. Research distribution of actuator technologies in AVs over the past five years, based on SCOPUS data. The value in parentheses indicates the quantity of related research projects.

Prototype implementations of these technologies include an AirGesture system, in which the driver has to remember a set of gestures [106], as well as the BullsEye system, which gathers relative positions of screen items [107].

Information detected using biometrics, such as a high cognitive workload, has to be communicated to the driver; one alternative is to use tactile displays, as drivers are able to confidently interpret important information, such as pedestrian status warnings and headway reduction signals when using higher urgency patterns, while still having short reaction time during a continuous driving task [108]. Regarding the visual architecture of the HMI system and its evaluation based on the driver's Decision-Making Time (DMT), there are some researches that study the DMT index when vehicles are expose to a risk situation in order to analyze the performance reliability of the HMI [109]. In addition, there is limited documentation of the Open Interfaces (OAS) focused on human-vehicle interactions, as the User Interface (UI) is very much based on smartphone use and not vehicle use; moreover, most of the cases interfaces reside in interoperability of the In-vehicle Information (IVI) with mobile devices [110].

E. SPECIFIC APPLICATION OF AUTONOMOUS VEHICLES HMIs

Even though there's active current research regarding HMI technologies for autonomous vehicles, there are some applications where these types of vehicles are being deployed. The use of these platforms serves as a precedent for experimenting and evaluating developing technologies in autonomous vehicles including interface design concepts.

- **Personal:** Currently, most of the personal use automated vehicles are at an SAE level of automation between 2 and 3, some examples include the Mercedes E Class, Tesla Model S, Volvo V90, BMW 7 series, and VW Passat, each with their own approach at monitoring and displaying internal and external information to their users through an interface [111]. Research regarding HMI design involves analyzing these platforms' interface design through the proposed transparency assessment method to estimate their understandability to identify critical elements of their HMIs that make users comfortable and knowledgeable of the systems [112], [113].
- **Commercial:** Autonomous vehicles have begun to provide commercial applications for users, one of these examples relies on using them for providing transportation services. Some examples include Uber cars with their XC90 and Fusion platforms and Google Car with their Prius platform, whose sensor setups offer autonomous transportation services with SAE level 4 automation capabilities, and their interface design have the objective of providing navigation and environmental information to their users [114]. Another commercial use of autonomous vehicles involves delivery applications, where delivery robots provide delivery services from a warehouse to a customer's living place. Research in HMI technologies for these robots involves the use of external interfaces for the robot to indicate its

navigational intent to pedestrians through text and lights, presenting information about its context [115].

- Military: Autonomous vehicles have many uses in military applications, including surveillance, inspection, logistics, and combat, among others which all need a network-based security application to integrate the required sources and services to function properly [116]. The design of proper interfaces for this application poses a challenge due to the increasing complexity and required tasks of the vehicles, which need much more computational resources not only for sensing and processing environmental information but also for coordination with additional vehicles that are deployed simultaneously. One example is the deployment of unmanned autonomous systems such as robotic platforms like UAVs, UUVs, or UGVs, whose human operators have had the challenge of using the real-time data obtained by these platforms to make decisions, and their interfaces should be able to anticipate disturbances while monitoring performance and responding to threats to ensure safety [117]. There is also a focus on designing interfaces to support the operation of as many complex UAVs as possible while reducing the number of operators needed, with the specific target of isolating cognitive processes and how sensory data is gathered within target environments [118].
- Agricultural: Multi-purpose autonomous vehicles are being deployed in agricultural environments to automate harvesting operations. Their adoption relies on their ability to be able to operate safely and avoid accidents, which is why research is focused on designing interfaces so that workers in the area can interact with the vehicle through the use of natural language, gestures, and portable devices for monitoring [119].

VI. DISCUSSION

Developments in automotive technology have led to current research in autonomous vehicles. One of the critical research areas focuses on the interaction between vehicles and users, which encompasses not only drivers and passengers but also external road users such as pedestrians and other vehicle drivers. These interactions aim to ensure safety and comfort by providing users with necessary information about the vehicle's status and its surroundings through the vehicle's interfaces.

Research in this area needs to be user-centered, addressing specific user needs inside and outside the vehicle, and considering external environmental elements that increase situational awareness. For instance, effective HMIs must provide clear and intuitive alerts to drivers during NDRTs to facilitate a smooth transition to manual control when necessary. These elements need to be communicated reliably through multiple stimuli, such as visual, auditory, and haptic feedback, to ensure a clear understanding of the situation. This approach enhances performance reliability and user acceptance, particularly with the increasing deployment of electric autonomous vehicles in urban areas.

The implementation of biometric technologies in HMIs is another significant area of development. These technologies monitor driver conditions during vehicle operation, generating alerts when needed and ensuring the decision-making process is not compromised by the vehicle's condition or external factors. For example, eye-tracking systems can detect driver drowsiness, while heart rate monitors can identify stress levels, ensuring timely alerts and interventions to maintain safety.

Currently, there are several challenges involved with the implementation of HMI technologies in autonomous vehicles, one of the most relevant is the lack of development of specific standardized protocols and regulations regarding these technologies. This lack of standards can cause difficulties and present obstacles to the implementation of developing technologies into a relatively new market such as autonomous vehicles. For instance, without standardized HMI protocols, a user familiar with one vehicle's interface might struggle with another, reducing confidence and slowing the adoption of autonomous vehicles. The development of standards tailored to HMIs for autonomous vehicles would help designers ensure consistency between designs so that users from different autonomous vehicle platforms can obtain familiar experiences within the vehicle. This would minimize the learning curve of interacting with the vehicle, such as the scenario where autonomous vehicles are used as a service for transportation by multiple companies, improving confidence and motivating the adoption of autonomous platforms. The creation of standards would also facilitate the interoperability between components of different manufacturers, which would allow for better integration and compatibility in this type of system. Standardization would also allow regulatory bodies to define minimum requirements and performance indexes for HMIs, reducing the risk of accidents and user errors.

Another relevant challenge lies in the availability of appropriate validation platforms for testing HMI designs. Researchers often resort to computer simulations and simulation cabins to test internal interfaces, which provide a controlled environment but may not reflect the complexities of real-life interactions. For example, simulation cabins like the VI-grade simulator can recreate the interior of an autonomous vehicle and project virtual test roads, but they might not capture the full range of user reactions seen in actual driving conditions. For external interfaces, virtual reality technology has been used to immerse test subjects in the role of pedestrians, providing better immersion and interactive environments. However, these simulations still have limitations in replicating real-life conditions dynamically.

Some researchers have managed to test external interfaces using physical validation platforms, ranging from simple autonomous vehicles to remotely controlled ones perceived as autonomous. These platforms allow for the evaluation of real-world interactions but come with significant economic and logistical challenges. They require substantial investment not only to acquire the platform but also to design infrastructure and implement safety measures. For instance, onfield testing platforms like those at Tecnológico de Monterrey involve high costs and complex coordination with authorities.

Addressing these challenges is crucial to ensure that HMI designs are thoroughly tested, validated, and capable of providing safe and effective driving experiences. Collaboration between researchers and industry, alongside the development of open-source platforms and standardization efforts, can help overcome these barriers and accelerate the advancement of HMI technologies in autonomous vehicles.

VII. CONCLUSION

The adoption of HMI technologies in autonomous vehicles offers numerous benefits that enhance their functionality and appeal in the market. Key benefits include enhanced safety through real-time feedback, alerts, and warnings, which mitigate risks and prevent accidents. Additionally, HMIs improve user experience by providing seamless access to vehicle status information, vehicle controls, entertainment systems, and personalized settings, which enhance comfort.

In terms of reliability, HMI systems in autonomous vehicles must incorporate key features such as redundancy and fault tolerance to ensure continuous operation despite sensor failures or software glitches. Additionally, user feedback mechanisms, usability testing, and validation processes should ensure that HMI systems meet user expectations and perform reliably in real-world scenarios.

Moreover, external HMIs play a critical role in communicating the vehicle's status and intentions to pedestrians and other road users, enhancing overall traffic safety. Effective external HMIs ensure that pedestrians receive clear and timely information about the vehicle's movements, reducing the likelihood of accidents and improving the interaction between autonomous vehicles and their environment.

Looking towards the future, several research directions and emerging trends in HMI design for autonomous vehicles need to be addressed. Future research should explore the integration of artificial intelligence and machine learning to create adaptive and predictive HMI systems that can anticipate user needs and preferences, taking also into account the external conditions in which the AV operates by identifying pedestrians and other vehicles' movements to act and communicate with them accordingly. Additionally, the development of more immersive and intuitive interfaces, such as those implementing augmented reality technologies, could significantly enhance user experience and engagement.

Research in HMI design for autonomous vehicles should also consider the integration of emerging technologies such as deep learning-based detection systems for identifying pedestrians, vehicles, and traffic conditions in complex urban environments which could improve AVs' ability to adapt to dynamic situations. Additionally, exploring realtime traffic management solutions using technologies that are embedded into the road infrastructure or external surveillance systems and predictive algorithms with the support of cloud computing will further address the challenges of navigating diverse environments. Finally, research should consider working towards standardizing HMI technologies across manufacturers to ensure an effective adoption in the market with a common direction.

Unanswered questions in the field include the long-term impact of HMI technologies on driver behavior and safety, the ethical implications of user data collection through the use of biometric technologies and its usage, and the potential for standardizing HMI designs across different vehicle manufacturers. Addressing these questions will be important for advancing the field and ensuring the successful adoption and acceptance of autonomous vehicles in the market.

Overall, the integration of advanced HMI systems in autonomous vehicles delivers a wide range of benefits, incentivizing the adoption and acceptance of autonomous vehicles. By addressing current challenges and focusing on future research directions, the field can continue to evolve and provide safer, more efficient, and more enjoyable driving experiences for all users.

REFERENCES

- H. U. Ahmed, Y. Huang, P. Lu, and R. Bridgelall, "Technology developments and impacts of connected and autonomous vehicles: An overview," *Smart Cities*, vol. 5, no. 1, pp. 382–404, Mar. 2022.
- [2] M. Martínez-Díaz and F. Soriguera, "Autonomous vehicles: Theoretical and practical challenges," *Transp. Res. Proc.*, vol. 33, pp. 275–282, Jan. 2018. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S2352146518302606
- [3] S. Debernard, C. Chauvin, R. Pokam, and S. Langlois, "Designing human-machine interface for autonomous vehicles," *IFAC-PapersOnLine*, vol. 49, no. 19, pp. 609–614, 2016. [Online]. Available: https://www.sciencedirect.com/science/article/ pii/S2405896316322418
- [4] C. Wilson, D. Gyi, A. Morris, R. Bateman, and H. Tanaka, "Nondriving related tasks and journey types for future autonomous vehicle owners," *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 85, pp. 150–160, Feb. 2022. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S1369847822000043
- [5] L. Shamseer, D. Moher, M. Clarke, D. Ghersi, A. Liberati, M. Petticrew, P. Shekelle, and L. A. Stewart, "Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015: Elaboration and explanation," *BMJ*, vol. 354, pp. 1–25, Jan. 2016. [Online]. Available: https://www.bmj.com/content/354/bmj.i4086
- [6] J. Carmona, C. Guindel, F. Garcia, and A. de la Escalera, "EHMI: Review and guidelines for deployment on autonomous vehicles," *Sensors*, vol. 21, no. 9, p. 2912, Apr. 2021. [Online]. Available: https://www.mdpi.com/1424-8220/21/9/2912
- [7] H. P. Singh and P. Kumar, "Developments in the human machine interface technologies and their applications: A review," *J. Med. Eng. Technol.*, vol. 45, no. 7, pp. 552–573, Oct. 2021, doi: 10.1080/03091902.2021.1936237.
- [8] Y. Li and J. Ibanez-Guzman, "LiDAR for autonomous driving: The principles, challenges, and trends for automotive LiDAR and perception systems," *IEEE Signal Process. Mag.*, vol. 37, no. 4, pp. 50–61, Jul. 2020.
- [9] M. Sahal, A. O. Kurniawan, and R. E. A. Kadir, "Object detection for autonomous vehicle using single camera with YOLOv4 and mapping algorithm," in *Proc. 4th Int. Seminar Res. Inf. Technol. Intell. Syst.* (ISRITI), Dec. 2021, pp. 144–149.
- [10] S. Lim, J. Jung, B.-h. Lee, J. Choi, and S.-C. Kim, "Radar sensor-based estimation of vehicle orientation for autonomous driving," *IEEE Sensors J.*, vol. 22, no. 22, pp. 21924–21932, Nov. 2022.

- [11] W. Xu, C. Yan, W. Jia, X. Ji, and J. Liu, "Analyzing and enhancing the security of ultrasonic sensors for autonomous vehicles," *IEEE Internet Things J.*, vol. 5, no. 6, pp. 5015–5029, Dec. 2018.
- [12] W. Rahiman and Z. Zainal, "An overview of development GPS navigation for autonomous car," in *Proc. IEEE 8th Conf. Ind. Electron. Appl.* (*ICIEA*), Jun. 2013, pp. 1112–1118.
- [13] X. Meng, H. Wang, and B. Liu, "A robust vehicle localization approach based on GNSS/IMU/DMI/LiDAR sensor fusion for autonomous vehicles," *Sensors*, vol. 17, no. 9, p. 2140, Sep. 2017. [Online]. Available: https://www.mdpi.com/1424-8220/17/9/2140
- [14] M. Baek, J. Mun, W. Kim, D. Choi, J. Yim, and S. Lee, "Driving environment perception based on the fusion of vehicular wireless communications and automotive remote sensors," *Sensors*, vol. 21, no. 5, p. 1860, Mar. 2021, doi: 10.3390/s21051860.
- [15] M. M. Madebo, C. M. Abdissa, L. N. Lemma, and D. S. Negash, "Robust tracking control for quadrotor UAV with external disturbances and uncertainties using neural network based MRAC," *IEEE Access*, vol. 12, pp. 36183–36201, 2024.
- [16] C. M. Abdissa, "Improved model predictive speed control of a PMSM via Laguerre functions," *Math. Problems Eng.*, vol. 2024, pp. 1–14, Mar. 2024. [Online]. Available: https://onlinelibrary. wiley.com/doi/abs/10.1155/2024/5562771
- [17] B. Derseh, L. Negash, and C. M. Abdissa, "Robust PSO tuned FOSMC for altitude stabilization and trajectory tracking of agricultural monitoring UAV," Addis Ababa Univ., Addis Ababa, Ethiopia, Tech. Rep. 24250348, Oct. 2023, doi: 10.36227/techrxiv.24250348.v1.
- [18] J. Wang, L. Zhang, Y. Huang, J. Zhao, and F. Bella, "Safety of autonomous vehicles," J. Adv. Transp., vol. 2020, pp. 1–13, 2020.
- [19] D. F. Araújo, Y.-F. Li, and H. Frölian, "A qualitative modeling for reliability and safety assessment of fully autonomous vehicle based on system-theoretic process analysis," in *Proc. 30th Eur. Saf. Rel. Conf. 15th Probabilistic Saf. Assessment Manage. Conf.*, 2020, pp. 84–91.
- [20] F. Wang, D. F. Araújo, and Y.-F. Li, "Reliability assessment of autonomous vehicles based on the safety control structure," *Proc. Inst. Mech. Eng., O, J. Risk Rel.*, vol. 237, no. 2, pp. 389–404, Apr. 2023, doi: 10.1177/1748006x211069705.
- [21] E. A. Teklu and C. M. Abdissa, "Genetic algorithm tuned super twisting sliding mode controller for suspension of maglev train with flexible track," *IEEE Access*, vol. 11, pp. 30955–30969, 2023.
- [22] W. Ayalew, M. Menebo, L. Negash, and C. M. Abdissa, "Solving optimal path planning problem of an intelligent mobile robot in dynamic environment using bidirectional rapidly-exploring random tree stardynamic window approach (BRRT*-DWA) with adaptive Monte Carlo localization (AMCL)," Addis Ababa Univ., Addis Ababa, Ethiopia, Tech. Rep. 24623784, Dec. 2023, doi: 10.36227/techrxiv.24623784.v1.
- [23] J. Zhou, L. Wang, and X. Wang, "Scalable evaluation methods for autonomous vehicles," *Exp. Syst. Appl.*, vol. 249, Sep. 2024, Art. no. 123603. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S0957417424004688
- [24] Society of Automotive Engineers. Taxonomy and Definitions for Terms Related to On-road Motor Vehicle Automated Driving Systems. Accessed: 25, Apr. 2023. [Online]. Available: https://www.sae. org/standards/content/j3016_202104
- [25] NHTSA. Car Safety Ratings. Accessed: Apr. 25, 2023. [Online]. Available: https://www.nhtsa.gov/ratings
- [26] AS Group. Vehicle Class Division. Accessed: Apr. 25, 2023. [Online]. Available: https://www.automotivescience.com/pages/vehicle-classdivision
- [27] K. Bengler, M. Rettenmaier, N. Fritz, and A. Feierle, "From HMI to HMIs: Towards an HMI framework for automated driving," *Information*, vol. 11, no. 2, p. 61, Jan. 2020. [Online]. Available: https://www.mdpi.com/2078-2489/11/2/61
- [28] K. Li, Y. Gong, and Z. Ren, "A fatigue driving detection algorithm based on facial multi-feature fusion," *IEEE Access*, vol. 8, pp. 101244–101259, 2020.
- [29] G. Meixner, C. Häcker, B. Decker, S. Gerlach, A. Hess, K. Holl, A. Klaus, D. Lüddecke, D. Mauser, M. Orfgen, M. Poguntke, N. Walter, and R. Zhang, *Retrospective and Future Automotive Infotainment Systems—* 100 Years of User Interface Evolution. Cham, Switzerland: Springer, 2017, pp. 3–53, doi: 10.1007/978-3-319-49448-7_1.
- [30] S.-J. Babak, S. A. Hussain, B. Karakas, and S. Cetin, "Control of autonomous ground vehicles: A brief technical review," *IOP Conf. Ser.*, *Mater. Sci. Eng.*, vol. 224, Jul. 2017, Art. no. 012029.

VOLUME 12, 2024

- [31] D. Barat, A. Fromion, S. Féron, L. Guen, and V. Lainé, "26–4: Invited paper: Automotive HMI: Present uses and future needs," in *SID Symp. Dig. Tech. Papers*, vol. 48, May 2017, pp. 374–376.
- [32] F. Ekman, M. Johansson, and J. Sochor, "Creating appropriate trust in automated vehicle systems: A framework for HMI design," *IEEE Trans. Hum.-Mach. Syst.*, vol. 48, no. 1, pp. 95–101, Feb. 2018.
- [33] S. Coeugnet, F. Mars, M. Bueno, C. Sentouh, J.-C. Popieul, A. Koustanaí, A. Pauzié, and H. Tattegrain, "A user-centered approach to adapt the human-machine cooperation strategy in autonomous driving," in *Proc.* 21st Congr. Int. Ergonom. Assoc. (IEA), N. L. Black, W. P. Neumann, and I. Noy, Eds., Cham, Switzerland: Springer, 2021, pp. 600–606.
- [34] National Highway Traffic Safety Administration, *Controls and Displays*, Standard 49 CFR 571.101, Federal Register, 2024, Accessed: Apr. 15, 2024. [Online]. Available: https://www.ecfr.gov/current/title-49/subtitle-B/chapter-V/part-571/subpart-B/section-571.101
- [35] NHTSA, Federal Motor Vehicle Safety Standards: Light Vehicle Brake Systems, Standard 135, Lab. Test Procedure, Dec. 2005, Accessed: Apr. 15, 2024. [Online]. Available: https://www.nhtsa.gov/lawsregulations
- [36] NHTSA. Federal Motor Vehicle Safety Standards: Electronic Stability Control Systems, Standard 126, Final Regulatory Impact Anal., Mar. 2007, Accessed: Apr. 15, 2024. [Online]. Available: https://www.nhtsa.gov/laws-regulations
- [37] International Organization for Standardization, *Road vehicles—* Functional Safety, Standard ISO 26262, 2018, Accessed: Apr. 15, 2024. [Online]. Available: https://www.iso.org/standard/68383.html
- [38] ISO, Road Vehicles—Ergonomic Aspects of Transport Information and Control Systems—Dialogue Management Principles and Compliance Procedures, Standard ISO 15005, 2017, Accessed: Apr. 15, 2024. [Online]. Available: https://www.iso.org/standard/69238.html
- [39] ISO, Ergonomics of Human-System Interaction—Part 112: Principles for the Presentation of Information, Standard ISO 9241-112, 2017, Accessed: Apr. 15, 2024. [Online]. Available: https://www.iso.org/standard/77520.html
- [40] SAE International, Information Report on Its In-Vehicle Message Priority, Standard SAE J2395, 2002, Accessed: Apr. 15, 2024. [Online]. Available: https://www.sae.org/standards/content/j2395_200202/
- [41] Y. Zhang, C. Yu, and Y. Shi, "Designing autonomous driving HMI system: Interaction need insight and design tool study," in *HCI International 2018—Posters' Extended Abstracts*, C. Stephanidis, Ed., Cham, Switzerland: Springer, 2018, pp. 426–433.
- [42] Y. Zheng and X. Ren, "Developing a multimodal HMI design framework for automotive wellness in autonomous vehicles," *Multimodal Technol. Interact.*, vol. 6, no. 9, p. 84, Sep. 2022. [Online]. Available: https://www.mdpi.com/2414-4088/6/9/84
- [43] N. Schömig, K. Wiedemann, S. Hergeth, Y. Forster, J. Muttart, A. Eriksson, D. Mitropoulos-Rundus, K. Grove, J. Krems, A. Keinath, A. Neukum, and F. Naujoks, "Checklist for expert evaluation of HMIs of automated vehicles—Discussions on its value and adaptions of the method within an expert workshop," *Information*, vol. 11, no. 4, p. 233, Apr. 2020. [Online]. Available: https://www.mdpi.com/2078-2489/11/4/233
- [44] F. Naujoks, K. Wiedemann, N. Schömig, S. Hergeth, and A. Keinath, "Towards guidelines and verification methods for automated vehicle HMIs," *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 60, pp. 121–136, Jan. 2019. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S1369847818301748
- [45] A. M. Tinga, D. Cleij, R. J. Jansen, S. van der Kint, and N. van Nes, "Human machine interface design for continuous support of mode awareness during automated driving: An online simulation," *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 87, pp. 102–119, May 2022. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S1369847822000614
- [46] X. Sun, S. Cao, and P. Tang, "Shaping driver-vehicle interaction in autonomous vehicles: How the new in-vehicle systems match the human needs," *Appl. Ergonom.*, vol. 90, Jan. 2021, Art. no. 103238. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S0003687020301885
- [47] F. Naujoks, Y. Forster, K. Wiedemann, and A. Neukum, "Improving usefulness of automated driving by lowering primary task interference through HMI design," *J. Adv. Transp.*, vol. 2017, pp. 1–12, Aug. 2017.

- [48] J. Chen, S. Wang, T. Zhou, L. Xiong, and X. Xing, "Study on safety analysis method for take-over system of autonomous vehicles," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Oct. 2020, pp. 1972–1977.
- [49] T. Gruden, S. Tomažič, J. Sodnik, and G. Jakus, "A user study of directional tactile and auditory user interfaces for take-over requests in conditionally automated vehicles," *Accident Anal. Prevention*, vol. 174, Sep. 2022, Art. no. 106766. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0001457522002020
- [50] F. Naujoks, Y. Forster, K. Wiedemann, and A. Neukum, A Human-Machine Interface for Cooperative Highly Automated Driving, vol. 484. Berlin, Germany: Springer, Jan. 2017, pp. 585–595.
- [51] Y. Forster, F. Naujoks, and A. Neukum, "Your turn or my turn: Design of a human-machine interface for conditional automation," in *Proc. 8th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* New York, NY, USA: Association for Computing Machinery, Oct. 2016, pp. 253–260, doi: 10.1145/3003715.3005463.
- [52] A. Eriksson, S. M. Petermeijer, M. Zimmermann, J. C. F. de Winter, K. J. Bengler, and N. A. Stanton, "Rolling out the red (and Green) carpet: Supporting driver decision making in automation-to-manual transitions," *IEEE Trans. Hum.-Mach. Syst.*, vol. 49, no. 1, pp. 20–31, Feb. 2019.
- [53] F. Schewe and M. Vollrath, "Visualizing the autonomous vehicle's maneuvers—Does an ecological interface help to increase the hedonic quality and safety?" *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 79, pp. 11–22, May 2021. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S1369847821000590
- [54] V. Papakostopoulos, D. Nathanael, E. Portouli, and A. Amditis, "Effect of external HMI for automated vehicles (AVs) on drivers' ability to infer the AV motion intention: A field experiment," *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 82, pp. 32–42, Oct. 2021. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1369847821001686
- [55] A. Orlický, A. Mashko, and J. Mík, "Assessment of external interface of autonomous vehicles," *Acta Polytechnica*, vol. 61, no. 6, pp. 733–739, Dec. 2021.
- [56] J. Guo, Q. Yuan, J. Yu, X. Chen, W. Yu, Q. Cheng, W. Wang, W. Luo, and X. Jiang, "External human-machine interfaces for autonomous vehicles from pedestrians' perspective: A survey study," *Sensors*, vol. 22, no. 9, p. 3339, Apr. 2022. [Online]. Available: https://www.mdpi.com/1424-8220/22/9/3339
- [57] N. Métayer and S. Coeugnet, "Improving the experience in the pedestrian's interaction with an autonomous vehicle: An ergonomic comparison of external HMI," *Appl. Ergonom.*, vol. 96, Oct. 2021, Art. no. 103478. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S0003687021001253
- [58] P. Bazilinskyy, L. Kooijman, D. Dodou, and J. C. F. de Winter, "How should external human-machine interfaces behave? Examining the effects of colour, position, message, activation distance, vehicle yielding, and visual distraction among 1,434 participants," *Appl. Ergonom.*, vol. 95, Sep. 2021, Art. no. 103450. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0003687021000971
- [59] A. Orlický, A. Mashko, and J. Mík, "Microsimulation model for assessment of eHMI of autonomous vehicles," in *Proc. Smart City Symp. Prague (SCSP)*, 2021, pp. 1–5.
- [60] S. M. Faas and M. Baumann, "Pedestrian assessment: Is displaying automated driving mode in self-driving vehicles as relevant as emitting an engine sound in electric vehicles?" *Appl. Ergonom.*, vol. 94, Jul. 2021, Art. no. 103425. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0003687021000727
- [61] K. Kalda, S.-L. Pizzagalli, R.-M. Soe, R. Sell, and M. Bellone, "Language of driving for autonomous vehicles," *Appl. Sci.*, vol. 12, no. 11, p. 5406, 2022. [Online]. Available: https://www.mdpi.com/2076-3417/12/11/5406
- [62] C. Ranasinghe, K. Holländer, R. Currano, D. Sirkin, D. Moore, S. Schneegass, and W. Ju, "Autonomous vehicle-pedestrian interaction across cultures: Towards designing better external human machine interfaces (eHMIs)," in *Proc. Extended Abstr. CHI Conf. Human Factors Comput. Syst.* New York, NY, USA: Association for Computing Machinery, Apr. 2020, pp. 1–8, doi: 10.1145/3334480.3382957.
- [63] M. Walch, K. Lange, M. Baumann, and M. Weber, "Autonomous driving: Investigating the feasibility of car-driver handover assistance," in *Proc. 7th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* New York, NY, USA: Association for Computing Machinery, Sep. 2015, pp. 11–18, doi: 10.1145/2799250.2799268.

- [64] S. Li, P. Blythe, W. Guo, A. Namdeo, S. Edwards, P. Goodman, and G. Hill, "Evaluation of the effects of age-friendly human-machine interfaces on the driver's takeover performance in highly automated vehicles," *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 67, pp. 78–100, Nov. 2019.
- [65] R. H. Y. Ma, A. Morris, P. Herriotts, and S. Birrell, "Investigating what level of visual information inspires trust in a user of a highly automated vehicle," *Appl. Ergonom.*, vol. 90, Jan. 2021, Art. no. 103272. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S0003687020302210
- [66] M. Walch, T. Sieber, P. Hock, M. Baumann, and M. Weber, "Towards cooperative driving: Involving the driver in an autonomous vehicle's decision making," in *Proc. 8th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.*, Oct. 2016, pp. 261–268.
- [67] VI-Grade. Compact Simulator. Accessed: Apr. 16, 2024. [Online]. Available: https://www.vi-grade.com/en/products/compact_simulator/
- [68] S. M. Faas, L.-A. Mathis, and M. Baumann, "External HMI for self-driving vehicles: Which information shall be displayed?" *Transp. Res. F, Traffic Psychol. Behaviour*, vol. 68, pp. 171–186, Jan. 2020. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S1369847819305479
- [69] C. G. Burns, L. Oliveira, P. Thomas, S. Iyer, and S. Birrell, "Pedestrian decision-making responses to external human-machine interface designs for autonomous vehicles," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2019, pp. 70–75.
- [70] C. Kettwich, R. Haus, G. Temme, and A. Schieben, "Validation of a HMI concept indicating the status of the traffic light signal in the context of automated driving in urban environment," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2016, pp. 1374–1379.
- [71] C. Olaverri-Monreal, S. Kumar, and A. Dìaz-Àlvarez, "Automated driving: Interactive automation control system to enhance situational awareness in conditional automation," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2018, pp. 1698–1703.
- [72] L. Davoli, M. Martalò, A. Cilfone, L. Belli, G. Ferrari, R. Presta, R. Montanari, M. Mengoni, L. Giraldi, E. G. Amparore, M. Botta, I. Drago, G. Carbonara, A. Castellano, and J. Plomp, "On driver behavior recognition for increased safety: A roadmap," *Safety*, vol. 6, no. 4, p. 55, Dec. 2020.
- [73] M. Anjum and S. Shahab, "Improving autonomous vehicle controls and quality using natural language processing-based input recognition model," *Sustainability*, vol. 15, no. 7, p. 5749, Mar. 2023. [Online]. Available: https://www.mdpi.com/2071-1050/15/7/5749
- [74] D. Fényes, B. Németh, and P. Gáspar, "A predictive control for autonomous vehicles using big data analysis," *IFAC-PapersOnLine*, vol. 52, no. 5, pp. 191–196, 2019. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2405896319306500
- [75] F. Faruffini, A. C. Victorino, and M.-H. Abel, "Vehicle autonomous navigation with context awareness," in *Proc. IEEE 2nd Int. Conf. Human-Machine Syst. (ICHMS)*, Sep. 2021, pp. 1–4.
- [76] M. Hasenjäger and H. Wersing, "Personalization in advanced driver assistance systems and autonomous vehicles: A review," in *Proc. IEEE* 20th Int. Conf. Intell. Transp. Syst. (ITSC), Oct. 2017, pp. 1–7.
- [77] S. Sibi, H. Ayaz, D. P. Kuhns, D. M. Sirkin, and W. Ju, "Monitoring driver cognitive load using functional near infrared spectroscopy in partially autonomous cars," in *Proc. IEEE Intell. Vehicles Symp. (IV)*, Jun. 2016, pp. 419–425.
- [78] F. Giobergia, E. Baralis, M. Camuglia, T. Cerquitelli, M. Mellia, A. Neri, D. Tricarico, and A. Tuninetti, "Mining sensor data for predictive maintenance in the automotive industry," in *Proc. IEEE 5th Int. Conf. Data Sci. Adv. Analytics (DSAA)*, Oct. 2018, pp. 351–360.
- [79] H.-G. Lee, D.-H. Kang, and D.-H. Kim, "Human-machine interaction in driving assistant systems for semi-autonomous driving vehicles," *Electronics*, vol. 10, no. 19, p. 2405, Oct. 2021. [Online]. Available: https://www.mdpi.com/2079-9292/10/19/2405
- [80] S. Iftikhar, Z. Zhang, M. Asim, A. Muthanna, A. Koucheryavy, and A. A. Abd El-Latif, "Deep learning-based pedestrian detection in autonomous vehicles: Substantial issues and challenges," *Electronics*, vol. 11, no. 21, p. 3551, Oct. 2022. [Online]. Available: https://www.mdpi.com/2079-9292/11/21/3551
- [81] M. Zohaib, M. Asim, and M. ELAffendi, "Enhancing emergency vehicle detection: A deep learning approach with multimodal fusion," *Mathematics*, vol. 12, no. 10, p. 1514, May 2024. [Online]. Available: https://www.mdpi.com/2227-7390/12/10/1514

- [82] H. Kawanaka, M. Miyaji, M. S. Bhuiyan, and K. Oguri, "Identification of cognitive distraction using physiological features for adaptive driving safety supporting system," *Int. J. Veh. Technol.*, vol. 2013, Jul. 2013, Art. no. 817179, doi: 10.1155/2013/817179.
- [83] L. Morra, F. Lamberti, F. G. Pratticó, S. L. Rosa, and P. Montuschi, "Building trust in autonomous vehicles: Role of virtual reality driving simulators in HMI design," *IEEE Trans. Veh. Technol.*, vol. 68, no. 10, pp. 9438–9450, Oct. 2019.
- [84] M. Z. Iskandarani, "Relating driver behaviour and response to messages through HMI in autonomous and connected vehicular environment," *Cogent Eng.*, vol. 9, no. 1, Dec. 2022, Art. no. 2002793, doi: 10.1080/23311916.2021.2002793.
- [85] F. A. Nazira, Md. I. Uddin, Md. H. Raju, S. Hossain, Md. N. Rahman, and M. F. Mridha, "Face recognition based driver detection system," in *Proc. Int. Conf. Data Analytics Bus. Ind. (ICDABI)*, Oct. 2021, pp. 572–577.
- [86] D. Vetturi, M. Tiboni, G. Maternini, and M. Bonera, "Use of eye tracking device to evaluate the driver's behaviour and the infrastructures quality in relation to road safety," *Transp. Res. Proc.*, vol. 45, pp. 587–595, Jan. 2020. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2352146520302155
- [87] T. Arakawa, "A review of heartbeat detection systems for automotive applications," *Sensors*, vol. 21, no. 18, p. 6112, Sep. 2021, doi: 10.3390/s21186112.
- [88] M. Deng, J. Chen, Y. Wu, S. Ma, H. Li, Z. Yang, and Y. Shen, "Using voice recognition to measure trust during interactions with automated vehicles," *Appl. Ergonom.*, vol. 116, Apr. 2024, Art. no. 104184. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S0003687023002223
- [89] S. Tateno, Y. Zhu, and F. Meng, "Hand gesture recognition system for in-car device control based on infrared array sensor," in *Proc. 58th Annu. Conf. Soc. Instrum. Control Engineers Jpn. (SICE)*, Sep. 2019, pp. 701–706.
- [90] M. Zhu, J. Chen, H. Li, F. Liang, L. Han, and Z. Zhang, "Vehicle driver drowsiness detection method using wearable EEG based on convolution neural network," *Neural Comput. Appl.*, vol. 33, no. 20, pp. 13965–13980, May 2021, doi: 10.1007/s00521-021-06038-y.
- [91] K. N. Minhad, S. H. M. Ali, J. O. S. Khai, and S. A. Ahmad, "Human emotion classifications for automotive driver using skin conductance response signal," in *Proc. Int. Conf. Adv. Electr., Electron. Syst. Eng.* (ICAEES), 2016, pp. 371–375.
- [92] J. F. Coughlin, B. Reimer, and B. Mehler, "Monitoring, managing, and motivating driver safety and well-being," *IEEE Pervasive Comput.*, vol. 10, no. 3, pp. 14–21, Jul. 2011.
- [93] M. A. Ramírez-Moreno, P. Carrillo-Tijerina, M. O. Candela-Leal, M. Alanis-Espinosa, J. C. Tudón-Martínez, A. Roman-Flores, R. A. Ramírez-Mendoza, and J. D. J. Lozoya-Santos, "Evaluation of a fast test based on biometric signals to assess mental fatigue at the workplace—A pilot study," *Int. J. Environ. Res. Public Health*, vol. 18, no. 22, p. 11891, Nov. 2021. [Online]. Available: https://www.mdpi.com/1660-4601/18/22/11891
- [94] J. M. Madrid, C. A. Arce-Lopera, and F. Lasso, "Biometric interface for driver's stress detection and awareness," in *Proc. 10th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* New York, NY, USA: Association for Computing Machinery, 2018, pp. 132–136, doi: 10.1145/3239092.3265970.
- [95] J. Bakker, M. Pechenizkiy, and N. Sidorova, "What's your current stress level? Detection of stress patterns from GSR sensor data," in *Proc. IEEE* 11th Int. Conf. Data Mining Workshops, 2011, pp. 573–580.
- [96] D. Yu, C. Park, H. Choi, D. Kim, and S.-H. Hwang, "Takeover safety analysis with driver monitoring systems and driver–vehicle interfaces in highly automated vehicles," *Appl. Sci.*, vol. 11, no. 15, p. 6685, Jul. 2021. [Online]. Available: https://www.mdpi.com/2076-3417/11/15/6685
- [97] G. Prabhakar and P. Biswas, "A brief survey on interactive automotive UI," *Transp. Eng.*, vol. 6, Dec. 2021, Art. no. 100089. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S2666691X21000452
- [98] X. Ge, X. Li, and W. Ying, "Methodologies for evaluating and optimizing multimodal human-machine-interface of autonomous vehicles," *SAE Tech. Paper*, vol. 2018, p. 12, Jan. 2018, doi: 10.4271/2018-01-0494.
- [99] K. S. Pečečnik, S. Tomažič, and J. Sodnik, "Design of head-up display interfaces for automated vehicles," *Int. J. Hum.-Comput. Stud.*, vol. 177, Sep. 2023, Art. no. 103060. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1071581923000691

- [100] P. Di Campli San Vito, E. Brown, S. Brewster, F. Pollick, S. Thompson, L. Skrypchuk, and A. Mouzakitis, "Haptic feedback for the transfer of control in autonomous vehicles," in *Proc. 12th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* New York, NY, USA: Association for Computing Machinery, Sep. 2020, pp. 34–37, doi: 10.1145/3409251.3411717.
- [101] D. Beattie, L. Baillie, M. Halvey, and R. McCall, "Maintaining a sense of control in autonomous vehicles via auditory feedback," in *Proc. 4th Int. Workshop Perceptual Quality Syst. (PQS)*, Sep. 2013, pp. 91–95.
- [102] S. I. N. Tientcheu, S. Du, and K. Djouani, "Review on haptic assistive driving systems based on drivers' steering-wheel operating behaviour," *Electronics*, vol. 11, no. 13, p. 2102, Jul. 2022. [Online]. Available: https://www.mdpi.com/2079-9292/11/13/2102
- [103] G. Ban and W. Park, "Effects of in-vehicle touchscreen location on driver task performance, eye gaze behavior, and workload during conditionally automated driving: Nondriving-related task and takeover," *Hum. Factors*, Jan. 2024, Art. no. 00187208241226838, doi: 10.1177/00187208241226838.
- [104] A. H. Shah and Y. Lin, "In-vehicle lighting and its application in automated vehicles: A survey," in *Proc. 5th Junior Conf. Lighting* (*Lighting*), Sep. 2020, pp. 1–4.
- [105] S. Cao, P. Tang, and X. Sun, "Driver take-over reaction in autonomous vehicles with rotatable seats," *Safety*, vol. 6, no. 3, p. 34, Jul. 2020. [Online]. Available: https://www.mdpi.com/2313-576X/6/3/34
- [106] K. R. May, T. M. Gable, and B. N. Walker, "A multimodal air gesture interface for in vehicle menu navigation," in *Adjunct Proc. 6th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* New York, NY, USA: Association for Computing Machinery, Sep. 2014, pp. 1–6, doi: 10.1145/2667239.2667280.
- [107] G. Weinberg, A. Knowles, and P. Langer, "Bullseye: An automotive touch interface that's always on target," in *Proc. 4th Int. Conf. Automot. User Interfaces Interact. Veh. Appl.* Portsmouth, NH, USA: Association for Computing Machinery, 2012, pp. 43–46. [Online]. Available: http://www.auto-ui.org/12/adjunct-proceedings/w1-02-weinberg.pdf
- [108] K. D. Martinez and G. Huang, "In-vehicle human machine interface: Investigating the effects of tactile displays on information presentation in automated vehicles," *IEEE Access*, vol. 10, pp. 94668–94676, 2022.
- [109] R. C. Goncalves, T. L. Louw, R. Madigan, M. Quaresma, R. Romano, and N. Merat, "The effect of information from dash-based humanmachine interfaces on drivers' gaze patterns and lane-change manoeuvres after conditionally automated driving," *Accident Anal. Prevention*, vol. 174, Sep. 2022, Art. no. 106726. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0001457522001622
- [110] T. Zieger and M. Kemppainen, "Human-machine interfaces are getting smarter," ATZelectronics Worldwide, vol. 16, no. 5, pp. 54–58, May 2021.
- [111] O. Carsten and M. H. Martens, "How can humans understand their automated cars? HMI principles, problems and solutions," *Cognition, Technol. Work*, vol. 21, no. 1, pp. 3–20, Feb. 2019, doi: 10.1007/s10111-018-0484-0.
- [112] Y.-C. Liu, N. Figalová, and K. Bengler, "Transparency assessment on level 2 automated vehicle HMIs," *Information*, vol. 13, no. 10, p. 489, Oct. 2022. [Online]. Available: https://www.mdpi.com/2078-2489/13/10/489
- [113] F. Naujoks, S. Hergeth, K. Wiedemann, N. Schoemig, and A. Keinath, "Use cases for assessing, testing, and validating the human machine interface of automated driving systems," in *Proc. Human Factors Ergonom. Soc. Annu. Meeting*, Oct. 2018, pp. 1873–1877.
- [114] E. Yurtsever, J. Lambert, A. Carballo, and K. Takeda, "A survey of autonomous driving: Common practices and emerging technologies," *IEEE Access*, vol. 8, pp. 58443–58469, 2020.
- [115] S. S. Kannan, A. Lee, and B.-C. Min, "External human-machine interface on delivery robots: Expression of navigation intent of the robot," in *Proc. 30th IEEE Int. Conf. Robot Human Interact. Commun.*, Aug. 2021, pp. 1305–1312.
- [116] R. Müller and C. Brook, "Net based waterside security applications: From small solutions to maritime security networks," in *Proc. Int. WaterSide Secur. Conf.*, Nov. 2010, pp. 1–6.
- [117] G. Vachtsevanos, B. Lee, S. Oh, and M. Balchanos, "Resilient design and operation of cyber physical systems with emphasis on unmanned autonomous systems," *J. Intell. Robotic Syst.*, vol. 91, no. 1, pp. 59–83, Jul. 2018, doi: 10.1007/s10846-018-0881-x.

- [118] J. Tweedale, "Using multi-agent systems to enhance the level of autonomy in unmanned vehicles," in Advanced Techniques for Knowledge Engineering and Innovative Applications, vol. 243. Berlin, Germany: Springer, Jan. 2012, pp. 189–208.
- [119] D. Kohanbash, M. Bergerman, K. M. Lewis, and S. J. Moorehead, "A safety architecture for autonomous agricultural vehicles," in *Proc. Amer. Society Agricult. Biol. Eng.*, Dallas, Texas, Jan. 2012, p. 1.



JESUS A. MANDUJANO-GRANILLO received the bachelor's degree in mechatronics engineering from Tecnológico de Monterrey, in 2021, where he is currently pursuing the M.Sc. degree in manufacturing systems. During his academic career, he worked on projects involving electronics, microcontrollers, embedded systems, and planar robotics, including mechanical design, and UI development. His research interests include interaction systems in autonomous vehicles, specifi-

cally focusing on HMI design and driving monitoring systems.



MILTON O. CANDELA-LEAL is currently pursuing the bachelor's degree in biomedical engineering with Tecnológico de Monterrey, Mexico. He conducted prior research at his university's NSF IUCRC BRAIN Center, creating machine learning models using wearables and biometrics (EEG, ECG, and computer vision) to predict cognitive states, such as mental fatigue, STEM interest, and emotion. He also collaborates with the FNNDSC at Boston Children's Hospital/Harvard

Medical School, developing deep-learning models with fetal MRI data for congenital heart disease classification and cortical structure segmentation.



JUAN J. ORTIZ-VAZQUEZ is currently pursuing the degree in mechatronics engineering with Tecnológico de Monterrey. Since 2021, he has been involved in robotics and aerospace projects that have received recognition in national and international competitions. His research interests include computer programming, vehicle automation, and aerodynamic designs.



MAURICIO A. RAMIREZ-MORENO (Member, IEEE) received the bachelor's degree in mechatronics engineering and the master's and Ph.D. degrees in biomedical engineering and physics, in 2013, 2015, and 2019, respectively. He is currently a Faculty Member of the International BRAIN Affiliate Site, Tecnológico de Monterrey, and a Candidate Member of the National Researchers System (SNI). He is also an Assistant Professor with the Department of Mechatronics,

Tecnológico de Monterrey, where he teaches biomedical engineering courses. His research interests include neuroengineering, machine learning applied to biomedical systems, biometry, and smart cities.



ALFREDO GALVAN-GALVAN is currently the Director of the School of Engineering and Sciences, Tecnológico de Monterrey, Monterrey. Previously, he was a National Business Development Officer and the Scientific Diplomacy Leader at Tecnológico de Monterrey. He contributed to the industrial sector as the President and the Vice President of the Nuevo León Software Council and the Leader of the Talent Development of the Nuevo León 4.0 initiative. He has more than 30 years

of professional and teaching experience in engineering and technological innovation projects related to the areas of education, automation, and design and development of mechatronic products.



JUAN C. TUDON-MARTINEZ (Member, IEEE) received the Ph.D. degree in sciences with a specialty in mechatronics from Tecnológico de Monterrey, in 2014, with three research stays in GIPSA-lab, INP Grenoble. From 2015 to 2021, he was an Associate Professor with Universidad de Monterrey. Since 2022, he has been a full-time Professor-Researcher with the School of Engineering and Science, Tecnológico de Monterrey. He is also a member of the National System of

Researchers in Mexico, level 1. He has been the director/co-director of ten postgraduate theses. During the last five years, he has led several consulting projects in Mexico, related to modeling and control theory applications (Vitro, Nutec-Bickley, Metalsa, and Michelin). He has published more than 80 research papers indexed in Scopus and/or JCR databases with an H-index of 18. His research interests include modeling and nonlinear control, vehicle dynamics, vehicle control, and autonomous vehicles.



LUIS C. FELIX-HERRAN is currently an Associate Professor with the School of Engineering and Sciences, Tecnológico de Monterrey. He is also a member of the Cyber-Physical Systems Research Group and Smart Mobility, Tecnológico de Monterrey. In addition to his teaching work in mechatronics and control systems, he conducts research on modeling/control of mechatronics focused on automotive suspensions, as well as renewable energies and electromobility.



JORGE DE J. LOZOYA-SANTOS received the Ph.D. degree in mechatronics and advanced materials from Tecnológico de Monterrey, in 2013, with research stays at GIPSA-Lab, INPG, Grenoble, France.

As a member of the National System of Researchers in Mexico Category 1, his work has made a significant impact. With over 49 international conferences, 53 indexed journals, one patent, and five patent applications, his contribu-

tions are undeniable. His research interests include intelligent transportation systems, system modeling and control, applied automatic control, and automotive systems are driving innovation at Tecnológico de Monterrey. He is the proud winner of the Tecnos Nuevo León 4.0 awards for his contributions to the state of Nuevo Leon, Mexico, in 2023 and 2024.

...